Author's response to reviews

Title: The feasibility and utility of grocery receipt analyses for dietary assessment

Authors:

Sarah L Martin (sjl2@cdc.gov)
Teresa Howell (t.howell@moreheadstate.edu)
Yan Duan (y-duan@morehead-st.edu)
Michele Walters (ma.walters@moreheadstate.edu)

Version: 2 Date: 28 February 2006

Author's response to reviews:

As suggested, I changed led to lead on page 5.
As suggested, I deleted text from the results section about age and race/ethnicity from the results section and instead, added similar text to the methods section on page 5.
In regard to the data abstracted and categorized from the receipts, there are interesting issues raised by the reviewer. In response, I added another limitation to the discussion.